Author's response to reviews

Title: Adherence to antiretroviral therapy in young children in Cape Town, South Africa, measured by medication return and caregiver self-report: a prospective cohort study.

Authors:

Mary-Ann Davies (Mary-Ann.Davies@uct.ac.za)
Andrew Boulle (Andrew.Boulle@uct.ac.za)
Tanzeeem Fakir (tanzeeem_faki23@yahoo.co.uk)
James Nuttall (James.Nuttall@uct.ac.za)
Brian Eley (Brian.Eley@uct.ac.za)

Version: 3 Date: 24 August 2008

Author's response to reviews: see over
Dear Editor

Final changes to MS: 1542571979182131

Authors: Mary-Ann Davies, Andrew Boulle, Tanzeem Fakir, James Nuttall and Brian Eley

Title: Adherence to antiretroviral therapy in young children in Cape Town, South Africa, measured by medication return and caregiver self-report: a prospective cohort study.

Herewith please find my revised submission to BMC Pediatrics with the final formatting changes and typographical corrections as requested in your email of Friday 15 August.

The following changes have been made:

1. Author affiliations have been changed to be written in full, in the requested format: department, institute, city, and country.

2. Table 3 is cited in the text in the last paragraph of the results section.

3. The word “commonest” has been replaced by “most common” in the fifth paragraph of the “Discussion” section as well as in the paragraph entitled “Experience with giving medication” in the “Results” section of the paper.

4. Underlined headings have been changed to bold.

5. Figure 2: The size of the font of the axis headings has been changed.

I request that BMC Pediatrics makes the remaining minor changes on my behalf.

Yours sincerely
Mary-Ann Davies